Author's response to reviews

Title: Mitigation of asthma triggers in indoor air: An inter-comparison of ventilation/filtration options for residences

Authors:

    Theodore A Myatt (tmyatt@eheinc.com)
    Taeko Minigishi (tminigishi@eheinc.com)
    Joseph G Allen (jallen@eheinc.com)
    David L MacIntosh (dmacintosh@eheinc.com)

Version: 2 Date: 14 February 2008

Author's response to reviews:

Hello Dr. Grandjean:

There is no real revision to the manuscript. I did not correctly include possible reviewers. Once submitted, it appears that you can not add this information into the website. Therefore I have listed 4 potential reviewers below. Sorry for the problems.

Cheers, Ted

The reviewers I would recommend include:

Dr. Richard Shaughnessy from the Indoor Air Quality Program at University of Tulsa
Rjstulsau@aol.com
(918) 230-3908

Dr. Robert Reisman from the University of Buffalo School of Medicine and Biomedical Sciences
reisman@buffalomedicalgroup.com 716.630.1130

Stuart Batterman from Environmental Health Sciences - University of Michigan
stuartb@umich.edu
734-763-2417

Qingyan (Yan) Chen from School of Mechanical Engineering Purdue University
yanchen@purdue.edu
(765) 496-7562